Author's response to reviews

Title: P53 nuclear stabilization is associated with FHIT loss and younger age of onset in squamous cell carcinoma of oral tongue

Authors:

Raju SR Adduri (sraju@cdfd.org.in)
Viswakalyan Kotapalli (kalyan@cdfd.org.in)
Neha A Gupta (guptaneha18@gmail.com)
Swarnalata Gowrishankar (swarnalata@gmail.com)
Mukta Srinivasulu (muktasrinivasulu@yahoo.co.in)
Mohammed M Ali (cuttosave@gmail.com)
Subramanyeshwar Rao (subramanyesh@gmail.com)
Shantveer G Uppin (drguppin@yahoo.co.in)
Umanath K Nayak (drumanathnayak@gmail.com)
Snehalatha Dhagam (doctorsneha@rediffmail.com)
Mohana V Chigurupati (drmohanavamsy@gmail.com)
Murali D Bashyam (bashyam@cdfd.org.in)

Version: 3
Date: 23 July 2014

Author's response to reviews: see over
July 23, 2014

Magdalena Morawska
Executive Editor
BMC Clinical Pathology

Sub: Submission of revised manuscript to BMC Clinical Pathology

Dear Dr. Morawska,

Thank you for reviewing our manuscript titled "P53 nuclear stabilization is associated with FHIT loss and younger age of onset in squamous cell carcinoma of oral tongue". We have now revised the manuscript based on queries/concerns raised by the editor. As per your request, I have attached a point by point response to editor’s comments. We hope that you will now find the manuscript acceptable for publication in your esteemed journal.

Thanking you,
Sincerely yours,

Murali Dharan Bashyam, PhD, MNAScI
Staff Scientist and Chief,
Laboratory of Molecular Oncology,
Centre for DNA Fingerprinting and Diagnostics (CDFD),
Tuljaguda complex, Nampally,
Hyderabad 500001, INDIA
Ph: 91-40-24749383
Fax: 91-40-24749448
E-mail: bashyam@cdfd.org.in
Response to editor’s comment

Comment 1:

Please upload the supplementary figures 1-3 as tif, jpeg or png.

**Response:** We thank the Editor for this advice. We have uploaded supplementary figures 1-3 in .tif format.